Author's response to reviews

Title: Differences in pregnancy outcomes and characteristics between insulin- and diet-treated women with gestational diabetes

Authors:

Katrien Benhalima (katrien.benhalima@uzleuven.be)
Karolien Robyns (karolien.robyn@uzleuven.be)
Paul Van Crombrugge (Paul.Van.Crombrugge@uzleuven.be)
Natascha Deprez (Natascha.Deprez@olvz-aalst.be)
Bruno Seynhave (Bruno.Seynhave@olvz-aalst.be)
Roland Devlieger (roland.devlieger@uzleuven.be)
Johan Verhaeghe (johan.verhaeghe@uzleuven.be)
Chantal Mathieu (chantal.mathieu@uzleuven.be)
Frank Nobels (Frank.Nobels@olvz-aalst.be)

Version: 4 Date: 9 October 2015

Author's response to reviews:

Dear Editor-in-Chief of BMC Pregnancy & Childbirth, 9-10-2015

We would like to submit the revised manuscript ‘Differences in pregnancy outcomes and characteristics between insulin- and diet-treated women with gestational diabetes’ for publication in BMC Pregnancy & Childbirth.

We have now added that ‘due to the retrospective nature of the study there was no need for informed consent from the participants as in compliance with the Belgian Law of December 8, 1992 on the protection of privacy and the Belgian Law of August 22, 2002 on the rights of the patient’. This has been highlighted in yellow in the methods section.